University of Wisconsin-Madison

CS/ECE 552: Introduction to Computer Architecture Spring 1999

Syllabus for Final Exam

(1) Pipelining: 6.7, 6.8. (2) Arithmetic: 4.6, 4.7, 4.8, 4.9, 4.10, carry save addition, other floating point material covered in class.

(3) Memory: 7.1, 7.2, 7.3, 7.4, 7.5, 7.6, 7.7 and error detection and error correction techniques (handout).

(4) Input/Output: 8.1, 8.3, 8.4, 8.5, 8.6, 8.7.

As always, you are responsible for all material that is covered in class.

Date and Location for the Examination

Saturday, May 8, 1999, 7:25-9:25 PM, in room 6102 Social Sciences.

It will be a closed book, closed notes examination.